Author's response to reviews

Title: A systematic literature review of pediculosis due to head lice in Pacific Island Countries and Territories: What country specific research on head lice is needed?

Authors:

Rick Speare (rickspeare@gmail.com)
Humpress Harrington (humpress.harrington@gmail.com)
Deon Canyon (dcanyon@hawaii.edu)
Peter D Massey (Peter.Massey@hnehealth.nsw.gov.au)

Version: 3  Date: 22 April 2014

Author's response to reviews:

1. a) Fig 1 has been cropped to remove the young girl's eyes. Neither of the two individuals can be identified.

   b) An additional description (East Kwaio Mountains) has been added to the location given originally in caption for Fig 1. We could add the name of the hamlet where the picture was taken, but since this is very small (population of approx 25), I am reluctant to do so for reasons of maintaining anonymity.

   c) I took the picture myself with permission from the chief.

2. A PRISMA checklist has been completed and uploaded.